Author’s response to reviews

Title: Prevalence and incidence of musculoskeletal extremity complaints in children and adolescents. A systematic review

Authors:

Signe Fuglkjaer (sfuglkjaer@health.sdu.dk)

Kristina Dissing (Kbdissing@health.sdu.dk)

Lise Hestbaek (l.hestbaek@nikkb.dk)

Version: 3 Date: 18 Sep 2017

Author’s response to reviews:

Dear Editor,

Thank you for the opportunity to submit a revised version of our manuscript; ID: BMSD-D-17-00271 about the prevalence and incidence of musculoskeletal extremity complaints in children and adolescents. Please, find below a response to each of the suggested revisions of the manuscript.

Yours sincerely,

Signe Fuglkjaer

1. Please upload all figures as separate figure files and remove them from the main manuscript. Please do not include the figure captions within the figure files, but instead place them at the end of the main manuscript in a section titled “Figure legends”.

ANSWER:

Figures are not included in the main manuscript, but uploaded as separate files and figure captions have been placed at the end of the main manuscript.

2. We would appreciate if you added a section titled “Supplementary files” where you list the following information about your supplementary material:
Please ensure that all additional files are explicitly referred to in the main text. Any items which do not meet these requirements may be deleted by our production department.

ANSWER: A section called supplementary files has now been added to the main manuscript (p. 16, line 363)

3. Upon uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

ANSWER: track changes have now been removed.